Author’s response to reviews

Title: Satisfaction of doctors with their training. Evidence from UK.

Authors:

Simon Gregory (sg641@cam.ac.uk)
Chiara Demartini (mariachiara.demartini@unipv.it)

Version: 4 Date: 07 Dec 2017

Author’s response to reviews:

1. Please add a “Conclusions” section after the “Discussion” section. This should state clearly the main conclusions of the research article and give a clear explanation of their importance and relevance.

We added a Conclusions section as required.

2. If you did not need formal ethics approval please confirm that this complies with national guidelines and provide a reference which supports this. Alternatively, supply a statement that says that a local ethics committee ruled that no formal ethics approval was required in this particular case.

We added a statement saying that we don't need ethical approval for this study according to national guidelines.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.